Author's response to reviews

Title: A systems biology approach to understand the pathophysiological mechanisms of cardiac pathological hypertrophy associated with rosiglitazone

Authors:

Lars Verschuren (lars.verschuren@tno.nl)
Peter Y Wielinga (peter.wielinga@tno.nl)
Thomas Kelder (thomas.kelder@tno.nl)
Marijana Radonjic (marijana.radonjic@tno.nl)
Kanita Salic (kanita.salic@tno.nl)
Robert Kleemann (robert.kleemann@tno.nl)
Ben van Ommen (ben.vanommen@tno.nl)
Teake Kooistra (teake.kooistra@tno.nl)

Version: 2
Date: 14 November 2013

Author's response to reviews:

Manuscript resubmission to BMC Medical Genomics

Dear Editor,

We have changed the formatting of our manuscript as indicated in your e-mail. The adjustments are described in a point-by-point description.

- We have added all email addresses of the authors in the title page.
- The format of the abstract is provided according to the guidelines.
- A heading for our conclusion is added in the manuscript body.
- The section “Competing interest” is added to the body of the manuscript.
- The section “Authors’ contributions” is added to the body of the manuscript.

We thank you for your help in processing this manuscript.

Sincerely,

Dr. Lars Verschuren